Author's response to reviews

**Title:** Outcomes following Percutaneous Coronary Intervention and Coronary Artery Bypass Grafting Surgery in Chinese, South Asian and White Patients with acute myocardial infarction: Administrative data analysis

**Authors:**

Danijela Gasevic (danijela.gasevic@fraserhealth.ca)
Nadia a Khan (nakhanubc@gmail.com)
Hong Quian (hqian@cheos.ubc.ca)
Shahzad Karim (Shahzad.Karim@fraserhealth.ca)
Gerald Simkus (Gerald.Simkus@fraserhealth.ca)
Hude Quan (hquan@ucalgary.ca)
Martha H Mackay (mmackay@providencehealth.bc.ca)
Blair J O'Neil (oneill2@ualberta.ca)
Amir F Ayyobi (Amir.Ayyobi@fraserhealth.ca)

**Version:** 3  **Date:** 13 March 2013

**Author's response to reviews:** see over
Dear BioMed Central Editorial Board,

Please consider our manuscript “Outcomes following percutaneous coronary intervention and coronary artery bypass grafting surgery in Chinese, South Asian and White patients with acute myocardial infarction (AMI)” for submission in BMC Cardiovascular Disorders.

We believe our manuscript will be of interest to your readership given that, to the best of our knowledge, no studies have explored the ethnic differences in outcomes in the post-AMI post-revascularization setting to date. Given that the burden of coronary heart disease has been steadily increasing in developing countries such as India and China, the use of coronary revascularization technology will likely rise globally. However, there is little evidence on the effectiveness of these procedures in South Asian and Chinese populations with AMI and how it compares to White populations. Several studies researching ethnic differences in outcomes in patients with coronary artery disease who have undergone revascularization procedures are available; however, no studies have focused on ethnic differences in outcomes following revascularization procedures exclusively in patients with AMI. Researching this population is important, as it is known that the effectiveness of revascularization differs between AMI populations and those with stable coronary artery disease.

All authors have read and approved the content of the revised manuscript. Thank you for your consideration.

Sincerely,

Nadia Khan

Dr. Nadia A. Khan
Centre for Health Evaluation and Outcome Sciences
1081 Burrard Street, room 620-B
Vancouver, BC, V6Z 1Y6
Telephone: 1-604-353-7548
Fax: 1-604-806-8005
Fax: 1-604-520-4799, Email: nakhanubc@gmail.com